

Pepperdine University Pepperdine Digital Commons

Theses and Dissertations

2020

Career trajectory and leadership: the role of personal adversity

Kristina E. Kovacs kkovacs21@gmail.com

Follow this and additional works at: https://digitalcommons.pepperdine.edu/etd



Part of the Business Commons, and the Organization Development Commons

Recommended Citation

Kovacs, Kristina E., "Career trajectory and leadership: the role of personal adversity" (2020). Theses and Dissertations. 1149.

https://digitalcommons.pepperdine.edu/etd/1149

This Thesis is brought to you for free and open access by Pepperdine Digital Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Pepperdine Digital Commons. For more information, please contact Katrina.Gallardo@pepperdine.edu, anna.speth@pepperdine.edu, linhgavin.do@pepperdine.edu.

CAREER TRAJECTORY AND LEADERSHIP: THE ROLE OF PERSONAL ADVERSITY

A Research Project

Presented to the Faculty of

The Graziadio Business School

Pepperdine University

In Partial Fulfillment of the Requirements for the Degree

Master of Science

In

Organization Development

by

Kristina E. Kovacs

July 2020

This research project, completed by

KRISTINA KOVACS

under the guidance of the Faculty Committee and approved by its members, has been

submitted to and accepted by the faculty of The Graziadio School of Business and

Management in partial fulfillment of the requirements for the degree of

MASTER OF SCIENCE IN ORGANIZATION DEVELOPMENT

Date: July 2020

Faculty Committee

Committee Chair, Darren Good, PhD

Committee Member, Christopher Worley, PhD

Deryck J. van Rensburg, Dean The Graziadio Business School

ii

Abstract

Adversity is part of life and certainly contributes to human development. This study seeks to explore the impact of personal adversity on career trajectory and leadership. A qualitative study based on 12 interviews with a narrative approach is presented. Key themes were identified among high-level executives around career evolution, perception of leadership, and how those experiences impacted their way of leading. The study concludes with a brief discussion of limitations and the findings that raise questions for additional research and provide some support for leadership and career development, counseling, and training.

Keywords: adversity, human development, career trajectory, leadership

Table of Contents

Abstract	iii
Table of Contents	
Chapter One: Introduction	1
Purpose Statement	5
Study Setting	5
Organization of the Study	5
Chapter Two: Review of Literature	7
Chapter Three: Methodology	23
Chapter Four: Results	26
Chapter Five: Discussion	42
References	48
Appendices	54

List of Tables

Table 1 Adversity	
Table 2 Career Trajectory: Evolution	31
Table 3 Career Trajectory: Ability to Reach this Level of Career	33
Table 4 <i>Leadership</i>	37
Table 5 Adversity: The Process and Ability to Grow from It	40

Chapter 1: Introduction

Media outlets tend to focus more heavily on leaders' positive qualities and daily routines as fundamental contributions to career success. To this day, psychologists have not sorted out which traits define leaders or if leadership exists outside of specific situations, and yet we know with absolute certainty that a handful of people have changed millions of lives and reshaped the world (Bennis, 2007). Previous research suggests leaders are challenged regularly by today's rapidly changing world, new technologies, globalization, the competitive landscape, scale up, and profitability (Powell & Baker, 2014).

Bennis and Thomas (2002) concluded that one of the most reliable indicators and predictors of true leadership is an individual's ability to find meaning in negative events and to learn from even the most trying circumstances. Through their interviews with top level executives, they highlighted examples of overcoming adversity. For example, Sidney Harman became a pioneer of participative management after workers rebelled in one of his factories. Liz Altman became a Motorola vice president after being transformed by the year spent at a Sony factory in rural Japan, where she faced estrangement and sexism. Muriel ("Mickie") Siebert, could not get a job as a stockbroker until she took her first name off of her resume and substituted a genderless initial; she then went on to become the first woman to own a seat on the New York Stock Exchange. While these stories are intriguing, there remains a lack of empirical research on the impacts of personal adversity experienced by leaders. Leadership literature has focused primarily on adversity or trauma experienced in the workplace as a consequence of the volatile nature of current affairs or the ever-changing competitive landscape of business,

while largely ignoring personal adversity. It is now commonly acknowledged that we bring our whole selves to work. People can use varying degrees of their selves (i.e., physically, cognitively, and emotionally) in the roles they perform, even as they maintain the integrity of the boundaries between who they are and the roles they occupy (Kahn, 1990). By further investigating the implications of personal adversity, we may discover that leveraging learnings from those experiences (e.g., an ill child, divorce, poverty, or death of a loved one) may have in fact primed us to tackle professional challenges.

Adversity refers to negative experiences that have the possibility to disturb a person's adaptive function or development (Yates & Masten, 2004). Research on adversity in individuals has produced a robust body of evidence consistently and, in some cases, prospectively, associating various predictors with actual individual resilient outcomes (Bonanno, Romero, & Klein, 2015). Adversity research has typically been focused on vulnerable populations, such as children, women, transgender, prisoners, etc. Research conducted by Mittal and colleagues (2015) indicates that adverse childhood environments do not universally impair mental functioning but can actually enhance specific types of cognitive performance in adults in the face of uncertainty.

We have not studied adversity experiences with leaders outside of the workplace and how those experiences may have impacted their career trajectory and leadership.

How and why have these individuals not only survived but thrived under adverse conditions? How have prior experiences with personal adversity provided insight or tools for the demands of their career?

This study investigates the role personal adversity may have in supporting the careers and leadership among high-level executives. Leadership is grounded in a

relationship. In its simplest form, it is a tripod—a leader or leaders, followers, and the common goal they want to achieve. None of those three elements can survive without the others (Bennis, 2007). Many people are exposed to loss or potentially traumatic events at some point in their lives, and yet they continue to have positive emotional experiences and show only minor and transient disruptions in their ability to function (Bonanno, 2004). Not everyone copes with these potentially disturbing events in the same way.

Some people experience acute distress from which they are unable to recover. Others suffer less intensely and for a much shorter period of time (Bonanno, 2004).

When faced with adversity people respond in a range of ways. Positive organizational scholarship "investigates positive deviance, or the ways in which organizations and their members flourish and prosper in especially favorable ways," (Cameron & Caza, 2004, p. 1). and "identifies the dynamics leading to exceptional individual organizational performance" (p. 1). Rather than draw the line between what is bad and what is good, the positive lens encourages scholars to explore what may be positive about seemingly neutral or even negative states, and how to transform conditions that are truly negative into those that are positive. Positive emotions and positive affect are intertwined and can be leveraged during times of personal adversity. Because positive emotions include a component of positive affect, positive emotions function as an internal signal to approach or continue (Fredrickson, 2001).

Resilience generally indicates how well an individual resists threats and how quickly they return to the initial state after a disturbance. In psychological research, resilience refers to individual strengths and can be defined as the ability of individuals to withstand stress and cope with pressure (Duchek, 2017). Research conducted by Bond

and Shaprio (2014) concluded that resilience was vital for career success — it is in the top three career success factors for our most senior interviewees. It is also learnable and should be an integral part of leadership development for the future.

Posttraumatic growth describes the experience of individuals whose development, at least in some areas, has surpassed what was present before the struggle with the crises occurred (Tedeschi & Calhoun, 2004). Posttraumatic growth is not simply a return to baseline-it is an experience of improvement that for some persons is deeply profound (Calhoun & Tedeschi, 2004). Prescod and Zeligman (2018) conducted a study among 215 undergraduate students on career adaptability of trauma survivors. They found a significant, positive relationship coupled with the predictive nature of intrusion symptoms, suggesting that experiencing trauma and negative trauma symptomatology might allow individuals greater career adaptability. Specifically, posttraumatic growth may buffer against negative trauma symptoms and might bring a sense of resiliency related to thriving in one's career.

Adversity is part of life and certainly contributes to human development. For over a decade, I have been the right-hand to C-suite leadership acting as an executive assistant and evolving to chief of staff. I have experienced, as a bystander, without full understanding the way adversity impacts leaders. I intend to address the gap in the research by exploring adversity among high-level leaders as it relates to personal experiences, leadership, and career trajectory. My hypothesis is that leaders are multifaceted and by learning how they may have made sense of their adversity and applied those experiences toward their careers and leadership, I will be able to add to the conversation of how we portray leaders. I will supplement leadership profiles beyond the

media's superficial routines by providing diverse examples for future leaders to identify with.

In addition, this study provides opportunities to add to research on leadership development and training programs and will have relevant application by identifying behaviors and/or mindsets for potential training techniques for current and future leaders, whether recent graduates entering the workforce or those looking to change or advance their career.

Purpose Statement

This study investigates the role personal adversity may have in supporting the careers and leadership among high-level executives including C-suite, Partner, President, or Founder. Three main research questions were examined:

- How have experiences with adversity affected your career trajectory?
- How have experiences with adversity affected you as a leader?
- To what do you attribute your ability to grow from adversity?

Study Setting

This study was conducted in offices, private meeting locations of the leaders' choosing, or virtually. The flexibility with location was to accommodate the participants' rigid schedules. The industries vary among the leaders interviewed, the range included legal, venture capital, consulting, ecommerce, entertainment, and retail. The strategy of inquiry for data collection were interviews with a narrative approach.

Organization of the Study

Chapter 1 provided the background and significance for the study including hypothesis, purpose, and study setting. Chapter 2 focuses on a deeper review of existing

literature on adversity, leadership, and different ways in which people respond to adversity including positive scholarship, positive emotions, resilience, and posttraumatic growth. Chapter 3 describes the methods used to conduct the present study including research design, procedures for recruiting participants, confidentiality, consent, data collection, and analyzing. Chapter 4 presents the findings of the study. Chapter 5 provides a discussion of the findings, including conclusions, recommendations, limitations, and directions for future research.

Chapter 2: Literature Review

This chapter provides a review of literature related to the study. The first body of literature reviewed is on adversity for the purpose of providing context for the study.

Next, leadership is reviewed with attention on the leader as an individual and the various behaviors, styles, and methods of leadership with a specific frame of reference in mind that current developments such as digitalization, globalization, and the omnipresence of crisis post great challenges for leaders. They must be able to anticipate potential threats, to cope effectively with critical situations, and to adapt to changing environments. Even more, they need to be able to use any changes for further development in order to grow despite crisis (Duchek, 2017). And, finally, various ways in which people have responded to adversity with a focus on positive scholarship including positive emotions, resilience, and posttraumatic growth.

Adversity

Adversity is part of life and certainly contributes to human development. Adversity refers to negative experiences that have the possibility to disturb a person's adaptive function or development (Yates & Masten, 2004). Adversities may be chronic (e.g., poverty, racism) or acute (e.g., sudden loss of a loved one, victim of an armed robbery). They may affect systems within the individual (e.g., a virus that attacks the immune system) or multiple levels and settings simultaneously (e.g., a natural disaster that affects individual systems of stress, beliefs, and behaviors, as well as broader systems of family, school, health care, agriculture, etc.) (Yates, Tyrell, & Masten, 2014).

Educational administrators see adversity as their constant companion. The conditions of adversity are clearer and more visible to administrators and teachers of

administration; they are implicit in such phrases as "declining enrollments," "diminished resources," "loss of confidence," and "accountability and assessment" (Culbertson, 2001).

Previous research indicates that adverse childhood environments do not universally impair mental functioning but can actually enhance specific types of cognitive performance in the face of uncertainty (Mittal, Griskevicius, Simpson, Sung, & Young, 2015). If, for example, a person grows up in an unpredictable and constantly changing environment, he or she ought to develop cognitive tendencies that help him or her function adaptively in this type of challenging environment. When tested in conditions of uncertainty (conditions reminiscent of their early life environments), adults exposed to more unpredictable childhood environments outperform those exposed to more predictable childhood environments on the executive function of shifting.

Rather than impairing cognitive functioning, Mittal et al.(2015) suggest people who had more unpredictable childhoods were worse at inhibition (overriding dominant responses), but better at shifting (efficiently switching between different tasks) when tested in conditions of uncertainty. Unpredictable environments, such as startups, where trends, new technology, and venture capital cash injections can cause unpredictable patterns of competition or vested interest, individuals "programmed" by unpredictable childhood experiences to identify new and better opportunities may be more likely to not only survive but thrive in these arduous environments.

Leadership

Today, the field of leadership focuses not only on the leader, but also on followers, peers, supervisors, work setting/context, and culture, including a much broader

array of individuals representing the entire spectrum of diversity, public, private, and notfor-profit organizations, and, increasingly over the past 20 years, samples of populations
from nations around the globe (Avolio, Walumbwa, & Weber, 2009). The focus of this
review is on the hardship's leaders experience, traits they exhibit, and coping
mechanisms they leverage both personally and professionally to create appropriate
context for the study. In the early 19th and 20th centuries, theories asserted that
leadership qualities were inherited, especially by people from upper class. "Great men"
were born, not made (Kirkpatrick & Locke, 1991).

The study of leader traits has a long and controversial history. While research shows that the possession of certain traits alone does not guarantee leadership success, there is evidence that effective leaders are different from other people in certain key respects (Kirkpatrick & Locke, 1991). Trait theories continue to be controversial since research showed that no traits were universally associated with effective leadership and situational factors were also influential.

Bennis and Thomas (2002) interviewed more than 40 leaders, young and old. They found that all of these leaders identified one very intense, transformational experience that they believed was critical to their leadership. Some of these personal "crucible" experiences were extremely positive (e.g., climbing a mountain), and others were tragic (e.g., loss of a child or being imprisoned for 16 years.) As a result of these crucible experiences, these leaders acquired a critical adaptive capacity that contributed to their success. Researchers define the extent of career adaptability as a reflection of an individual's self-concept, perceptions about life, anxious response, and mood, which are

also affected by employment stability (Maggiori, Johnston, Krings, Massoudi, & Rossier, 2013).

To perform effectively in complex mission environments, security personnel and leaders must be flexible and adaptable in responding to rapidly changing conditions. Psychological hardiness marks resilient people who maintain their health and performance despite stressful situations (Barton, Kelly, & Matthews, 2013). People define career resilience both as an in-the-moment response to set-backs and crisis, and as a long-term behavior. Four out of five people define resilience as "recovering well from set-backs" (Bond & Shaprio, 2014, p. 4), but 64% also said it is about the ability to adapt well to change, and 61% said it is about being tenacious/keeping on going.

Although there is a lack of empirical research, the transformational leadership literature suggests that exposure to trauma has both positive and negative impacts on leadership and leadership development. Transformational leaders stimulate thought by soliciting input of others, encouraging followers to challenge old ways of operation, view problems from a new perspective, participate in developing new, more efficient work processes, and overcome resistance to change (McClellan, Levitt, & DiClementi, 2017).

The ability to lead is related to the emotional, cognitive, and physical well-being of the leader. Trauma, whether witnessed or resulting in personal injury, has the potential to alter leaders' wellbeing and change their outward leadership behaviors (Kramer & Allen, 2018). Leaders are obviously human beings with the full range of moods and emotions potentially available to them. Both positive and negative moods and emotions serve numerous functions in people's lives. Likewise, both positive and negative moods and emotions can sometimes be the cause of human dysfunction, and therefore, emotional

intelligence may be a key contributor to leadership effectiveness (George, 2000). For example, a leader who is surprised when fear and anxiety are followers' initial reaction to an announced restructuring (even with a guarantee of no layoffs) is not knowledgeable about the determinants of emotions (George, 2000).

Prior empirical studies have confirmed the assertion that employees' attitudinal and behavioral reactions to change play a major role in its success. Shin, Taylor, and Seo's (2012) research on organizational change indicates managers who are concerned about their employees' commitment to change should consider psychological resilience as one criterion for the selection of new employees and as content for training interventions.

Leaders need to manage emotions such that followers are aware of problems yet, given the collective vision, are confident about resolving problems and feel optimistic about the efficacy of their personal contributions (George, 2000). In organizational studies, there has been a growing interest in applying the principles and methodology of positive scholarship to micro and macro-level organizational issues. Within organizational studies, the positive perspective has sharpened the focus on positive states (e.g., authenticity, optimal performance, engagement, thriving, high-quality connections, social responsibility, sustained peace, dynamic capabilities) and their generative mechanisms (e.g., empowerment, trust, creativity, humanistic work ideology) (Roberts, 2006).

The leadership literature, like the positive psychology literature, indicates that positively making meaning of life's difficult moments contributes in positive ways to one's life and to the impact that life will have on an organization. Conflict researchers

have mainly focused on task conflict and negative emotions. Todorova, Weingart, and Bear (2013) drew on affect events theory which suggests that events (such as conflicts) can have either positive or negative effects on peoples' emotions depending on whether they appraise them positively or negatively. Todorova et al. (2013) suggest that task conflict expressed with mild intensity leads to positive, energizing emotions. For example, founders who defined a situation as an opportunity simultaneously embraced the adversity, those who defined it as a challenge sought to counter the adversity, and those who defined a situation as a threat attempted to accommodate the adversity (Powell & Baker, 2014).

The leadership literature continues to evolve as the world becomes more complex, therefore, it is necessary to take a more holistic view of leadership. Researchers are now examining all angles of leadership, including models and studies of the leader, the follower, the context, the levels, and their dynamic interaction as well as the process of leadership (e.g., integrating cognitive psychology with strategic leadership) (Avolio, Walumbwa, & Weber, 2009). Research to date on adversity has been limited by focused studies on industry or population. This study investigates how high-level leaders may have made sense of their adversity, grown from it, and now apply it to their leadership.

Responding to Adversity

When faced with adversity, people respond in a range of ways. For the purpose of this study I have chosen to focus on positive scholarship including positive emotions, resilience, and post-traumatic growth.

Positive scholarship including positive emotions. A growing number of researchers have helped us understand that organizations are most vibrant and alive when

they embrace the tensions of the human condition (Bright, 2009). Rather than focus on the paradoxical dark side of positive scholarship alluded to by Roberts (2006), as an elusive ploy to exploit the powerless through rhetoric of positive scholarship to drive people to work harder, give more, and commit fully, for the purpose of this study and its focus on high-level leaders' experiences with personal adversity and the impact it has had on them professionally, I will focus on the more humble side, that positive scholars seek to ignite and nurture an individual's potential for intrinsic, positive valuations, even though the goal may be external to the actor (Fineman, 2006).

Positive organizational scholarship is an area of study that categorizes previous research and provides an organizing frame for current and future research on positive states, outcomes, and generative mechanics in individuals, dyads, groups, organizations, and societies (Roberts, 2006). Building extraordinary organizations is not a matter of focusing exclusively on the positive; it is a matter of understanding how so-called "positive forces" (e.g., creativity, innovation, positive emotions, etc.) function in dynamic relationship with so-called "negative forces" (e.g., negative emotions, conflict, etc.). Healthy organizations are not characterized by an absence of negativity; rather, they generate a nurturing climate in which experiences of all colors from positive to negative can be harnessed to sustain and perpetuate flourishing organization life (Bright, Powley, Fry, & Barrett, 2013).

The initial intent of positive scholarship was to discover the mechanisms that enable human flourishing; the call for positive scholarship was grounded in an implicit desire to enhance the quality of life for individuals who work within and are affected by organizations (Roberts, 2006). Positive emotions and positive affect are intertwined and

can be leveraged during times of personal adversity. Because positive emotions include a component of positive affect, they too function as internal signals to approach or continue. Even so, positive emotions share this function with a range of other positive affective states (Fredrickson, 2001).

Positive emotions arise in response to diffuse opportunities rather than narrowly focused threats. Positive emotions momentarily broaden people's attention and thinking, enabling them to draw on higher-level connections and a wider-than-usual range of percepts or ideas (Fredrickson, Coffey, Pek, Cohn, & Finkel, 2008). Researchers (Fredrickson, 2001; Fredrickson & Losada, 2005; Fredrickson, Tugade, Waugh, & Larkin, 2003; McClellan, Levitt, & DiClementi, 2017) see positive emotions as active ingredients in superior coping and thriving despite adversity. Studies suggest that in moments of crisis positive emotions do more than feel good in the moment. Fredrickson et al. (2003) studied resilience and emotions following the terrorist attacks on the United States on September 11th, 2001. They suggest that through experiences of positive emotions, people may literally transform themselves, becoming more creative, knowledgeable, socially integrated, healthy, and resilient individuals.

Drawing on advocates of emotional intelligence, Mayer and Salovey (1993) defined emotional intelligence:

A type of social intelligence that involves the ability to monitor one's own and others' emotions, to discriminate among them, and to use the information to guide one's thinking and actions. The scope of emotional intelligence includes the verbal and nonverbal appraisal and expression of emotion, the regulation of

emotion in the self and others, and controversy the utilization of emotional content in problem solving. (p. 433)

Leaders use emotion to increase their ability to motivate, inspire, and influence followers in an interpersonal manner that takes into consideration both the need to overcome negative emotions as well as the power associated with promoting positive emotions (McClellan, Levitt, & DiClementi, 2017). Research has linked positive moods to creativity that suggests when leaders are in positive moods, they may be more creative and more likely to come up with a compelling vision that contrasts with the existing conditions. Therefore, it may be possible that such leaders are also likely to be better able to repair negative moods arising from any number of sources that may limit flexibility. Therefore, a case could be made that leaders who have experienced adverse circumstances may draw upon positive emotions for creative roadmaps toward balance or potential growth from trauma (George, 2000).

Resilience. Although the term resilience has been in broad use for centuries, it was only in the past several decades that it gained currency as a psychological construct (Bonanno, Romero, & Klein, 2015). Resilience is usually considered to be an ability to go on with life after hardship and adversity, or to continue living a purposeful life after experiencing hardship and adversity (Calhoun & Tedeschi, 2004). Resilience, whether it is defined in terms of capacity, processes, or outcomes of positive adaptation in contexts of risks, will depend on the coaction of multiple systems as they come together in the function or development of the individual (Masten, 2015).

Resilience appears to be a common phenomenon arising from ordinary human adaptive processes. Human adaptation and development are shaped by many interactions

across levels, ranging from the molecular level to social and ecological systems. Therefore the development of a child or adult, including that person's potential or manifested resilience, will be influenced by many interactions within the individual (e.g., genetic, neural, immunological, cognitive) and also between the person and the environment, including interactions with family, peers, school, community, and the natural and built environment (Masten, 2015). Resilience is a superordinate construct subsuming two distinct dimensions -- significant adversity and positive adaptation -- and thus is never directly measured, but rather is indirectly inferred based on evidence of the two subsumed constructs (Luthar, 2006).

Prior research from Bonanno, Romero, and Klein (2015) discuss the evolution of resilience and its distinction between acute and chronic circumstances. In the general sense, the category of acute adversity describes a relatively isolated but potentially traumatic life event that demands resources and/ results in the loss of resources and exerts its primary impact over a relatively transient period, usually no longer than one month. By contrast, chronic adversity involves an event or related series of events that exerts repeated and cumulative impact on resources and adaptation and persists for many months and typically considerably longer (Bonanno, Romero, & Klein, 2015).

Individuals are not considered resilient if there has never been a significant threat to their development; there must be current or past hazards judged to have the potential to derail normative development (Masten, 2001). Resilience has been studied across various adverse circumstances from refugees, former child soldiers, and bereavement, to those suffering from chronic distress, recurrent intrusive memories, or sadness from years after exposure to an adverse event.

Resilience manifests in different ways. A study of Cambodian refugee survivors of the Khmer Rouge period found the resilient did not lose the consciousness of their participation or their responsibility for their own lives and destinies (Overland, 2011). They refused to become victims and reaffirmed their self-reliance both by working, by fighting for freedom, and by struggling not to give up. For those coping with the death of a spouse, resilient people were better able than less resilient participants to gain comfort from talking or thinking about the spouse, reported the fewest regrets about their behavior with the spouse, and reported fewer things they may have done or failed to do when they were still alive. Resilient individuals were less likely to search in order to make sense of or find meaning in the spouse's death (Mancini & Bonanno, 2006). Bond and Shapiro (2014) conducted a study on women, resilience, and their career success and found that 76% of people at Board level said that resilience is essential to career success, but only 10% of people at any level say that their organization placed a lot of emphasis on building and maintaining resilience as a factor in career success

Time is essential in thinking about resilience because adaptation unfolds over it.

The patterning of adaptation through time is shaped by many influences, all of which can interact and change. These include the patterns of challenge or adversity exposure in time, fluctuating functions of the individual at many levels that could function to moderate responses to challenges, the ebb and flow of resources and relationships that could support adaptation at many levels, and change in the contexts of life, including many aspects of home, family, school, community, culture, and media (Masten, 2015).

The great surprise of resilience research is the ordinariness of the phenomena.

Resilience appears to be a common phenomenon that results in most cases from the

operation of basic human adaptational systems (Masten, 2001). The survival of an organization during crisis is dependent on the resilience of its members as well as its leadership (Teo, Lee, & Lim, 2017). Teo et al. (2017) suggest that one of the primordial tasks of leaders in a crisis is to recognize the early signs, invoke that a threshold has been crossed, and usher the organization into a new phase of the organizational lifecycle where new routines and structural patterns can be learned. Previous studies highlighted that resilient employees are better prepared to overcome difficulties and stressful events and are able to find a positive meaning in negative circumstance, becoming more adaptive and successful over time (Tugade & Fredrickson, 2004; Youssef & Luthans, 2007).

Posttraumatic growth. Tedeschi and Calhoun (1996, 2004) define posttraumatic growth as the experience of positive change that occurs as a result of the struggle with highly challenging life crises and identified five major domains, including greater appreciation of life and changed sense of priorities; warmer, more intimate relationships with others; a greater sense of personal strength; recognition of new possibilities or paths for one's life; and spiritual development.

Calhoun and Tedeschi (2004) provide a framework for understanding the process of posttraumatic growth with an emphasis on the role of cognitive processing. Cognitive rebuilding takes into account the changed reality of one's life after trauma produces schemas that incorporate the trauma and possible events in the future, and that are more resistant to being shattered. These results are experienced as growth. Growth, however, does not occur as a direct result of trauma. It is the individual's struggle with the new reality in the aftermath of trauma that is crucial in determining the extent to which posttraumatic growth occurs. It is also suggested that posttraumatic growth mutually

interacts with life wisdom and the development of the life narrative, and that it is an ongoing process, not a static outcome (Calhoun & Tedeschi, 2004).

Aldwin and Levenson (2004) argue that Tedeschi and Calhoun's (2004) posttraumatic growth measures assess only positive outcomes, which does not permit contrast with negative outcomes; stressors and positive events may promote development in adulthood that is not restricted to traumatic events. Many researchers agree that the evidence seems to support the contention that it is not so much the event but how individuals cope with it that determines positive versus negative outcomes. Further, emotion regulation is an important component, as is cognitive processing (Aldwin & Levinson, 2004; Calhoun & Tedeschi, 1999, 2004). However, similar to the argument from Aldwin and Levinson (2004) that suggests Tedeschi and Calhoun (2004) do not devote adequate attention to coping strategies as mediators of the relation of stress and development outcomes, Janoff-Bulaman (2004) provides a more detailed explanation of how coping processes helps account for posttraumatic growth.

Successful coping or recovery from trauma does not mean returning to one's earlier fundamental assumptions, but rather establishing a comfortable, integrated assumptive world that incorporates the traumatic experience. The trauma is permanently encoded in the survivor's psyche via changes in these basic schemas that reflect some degree of both disillusionment and personal vulnerability; that is, at the level of their fundamental assumptions, survivors are left with somewhat more negative views of the world and their own security (Janoff-Bulman, 2004). As a result, Janoff-Bulman (2004) offers three distinct models of posttraumatic growth: strength through suffering,

psychological preparedness, and existential reevaluation that account for the positive and negative being inextricably linked. The survivor can focus on one or the other.

Posttraumatic growth manifests in different ways. A study on posttraumatic stress and posttraumatic growth among primary breast cancer patients reported enhanced appreciation of life and interpersonal relationships were most salient, whereas new opportunities were least frequently identified (Cordova, Giese-Davis, Golant, Kronenwetter, Change, & Spiegel, 2007). Posttraumatic growth is not simply a function of socially desirable reporting in which patients uniformly and globally rate themselves as having changed for the better. Rather, patients appear to have a differentially greater awareness of the value of life and relationships (Cordova et al., 2007).

Tedeschi (2011) leverages Janoff-Bulman's (2004) models to facilitate PTG (posttraumatic growth) as an intervention through understanding trauma response as a precursor to growth, enhancing emotional regulation, constructive self-disclosure, creating a trauma narrative with PTG domains, and developing life principles that are robust to challenges (Tedeschi, 2011). Combat veterans and their families face significant challenges not only to their ability to cope, but often to their fundamental belief systems (Tedeschi, 2011). Transforming the doubt, guilt, and pain of posttrauma living into a clear sense of direction involves arriving at a set of principles that serve to guide decisions and actions to meet future challenges, thus promoting resilience (Tedeschi, 2011).

Znoj (2005) studied bereavement and posttraumatic growth with similar findings for the important variables for PTG including the importance of emotional regulation, acceptance, and taking a positive stance. As life with its daily chores and activities goes

on, people might get used to their emotional states and could learn tolerating emotionally ambiguous situations better than without the experience of loss (Znoj, 2005).

As Calhoun and Tedeschi (2004) suggest, it was not until the 1980s, and then more strongly in the 1990s, that systematic scholarly interest specifically focused on the possibility of growth from the struggle with trauma emerged. Investigations in this area can inform us about psychological phenomena about which we know very little, and as they do so, they can provide significant information for those who attempt to provide assistance to those coping with major life disruptions (Calhoun & Tedeschi, 2004). Even as we examine those aspects that are amenable to the methodologies now at hand, some analysts must be willing to look at leadership in all its complexity, which may mean looking at elements that cannot be nailed down in the laboratory (Bennis, 2007).

Posttraumatic growth is both a process and an outcome. As research states there are additional elements to be considered, I intend to explore the experiences of leaders who not only bounce back from adversity which could include trauma, but use it as a springboard to further individual development or growth, and the development of more humane social behaviors and social organizations (Tedeschi, Park, & Calhoun, 1998).

Conclusion

The need for leadership development has never been more urgent; the implementation of ongoing development and training may have not only positive professional but personal outcomes. "Lifelong learning" has been a buzzword in corporate and university circles for decades, but it is still far from a reality. Companies of all sorts realize that to survive in today's volatile, uncertain, complex, and ambiguous environment, they need leadership skills and organizational capabilities different from

those that helped them succeed in the past. There is also a growing recognition that leadership development should not be restricted to the few who are in or close to the C-suite (Moldoveanu & Narayandas, 2019).

Organizations may also implement strategies to promote the development of problem-solving behaviors in the work setting, such as developing training sessions on ways to cope with adversity or uncertainty through focus groups or implementing suggestion schemes. When problem-solving behaviors are part of everyday organizational life, employees are better prepared to respond to setbacks and crises when they happen (Dimas, Rebelo, Lourenco, & Pessoa, 2018). Research on naturally occurring resilience suggested that there were windows of opportunity for intervention, when developmental processes, contexts, and opportunities converged to support positive change and a high return on investment. One example is the transition to adulthood, a transition when brain development, motivation, mentoring, training, and other opportunities appear to support positive redirection of the life course (Masten, 2015).

This study aims to investigate the way leaders have experienced personal adversity, have grown from it, and how they may have leveraged those experiences to flourish in their careers and lead others. The intent is to take these learnings and add to the conversation of how we portray leaders beyond the media's superficial routines by providing diverse examples for future leaders to identify with. In addition, this study intends to add to the existing literature on leadership and training programs that aid in the development of high-level leaders so others may benefit from those learnings.

Chapter 3: Methodology

The purpose of this study was to understand personal experiences with adversity among high-level leaders and how they may have drawn on those experiences to support their careers and leadership.

Sample

Participants were identified via the Principal Investigator's professional and personal network. Request for participation was made via email with consideration of the criteria: high-level executives, including C-suite, Partner, Founder, or President. 12 interviews were conducted in total including seven men and five women. Ages ranged from 35-55 and experience as a high-level executive fluctuated between 10-20 years. The industries varied among leaders interviewed, including legal, venture capital, consulting, ecommerce, entertainment, and retail.

Research Relationship

Although I have a personal relationship with some of the participants, I entered the relationship as a researcher with a career built on supporting C-suite executives. Participants were asked to consider the questions as an opportunity to have a conversation for a more friendly and trusting tone rather than researcher/participant, which may be viewed as more intimidating.

To ensure the privacy of participants, the following safeguards were employed to protect their rights:

• Each participant was provided an Informed Consent form via email with request to participate (Appendix A). The research objectives were articulated verbally, and consent was attainted prior to interview.

23

- The participants were informed of all data collection devices and activities.
- Participants were assured that they would remain anonymous; there were no reference to names or organizations.
- Participants were provided access to verbatim transcriptions and written interpretations/reports if interested.

Data Collection

Data was collected through face-to-face or one-on-one video conference interviews lasting no more than 60 minutes using a multi-question script with a narrative approach (Appendix B). Each interview was recorded, transcribed, coded, and analyzed. Questions asked were thought provoking to better understand the role personal adversity may have had in supporting the participant's careers and leadership.

Research Design

This study employed a qualitative interviewing design. Qualitative methods allow a depth of inquiry to occur during the course of the study (Creswell, 2013). A drawback of qualitative research approaches is the researcher bias that can affect the collection and analysis of results. For example, it is important for researchers to understand that what questions they ask participants and how they ask those questions can lead the participant. Therefore, it is important to pay attention to the question wording and phrasing as well as participants' responses. In addition, the participants were sourced from the researcher's network which may introduce additional bias and varying levels of comfortability among participants.

Interviewing poses strengths and limitations. Its benefits are that intangible data such as nonverbal language can be captured, and feelings and thoughts can be probed in

depth. Additionally, interviews can capture process info that surveys cannot. The primary challenge of research interviews is that they often produce a tremendous volume of information that can be difficult to analyze, absorb, and interpret (Creswell, 2013).

The qualitative interviewing approach was considered appropriate for this study due to the lack of in-depth literature about leadership styles and executive coaching among leaders in the entertainment industry. Qualitative research has been identified as an appropriate method in such cases (Creswell, 2013). Also, the interview method would allow me to gain a depth of insight about leaders' perceptions, attitudes, and interpretations of the topics being examined. This kind of information would be difficult to gain through other methods such as survey, observation, or archival research.

Data Analysis

This study used a categorical coding matrix, data was organized categorically and chronologically, reviewed repeatedly, and continually coded. A list of major themes that surfaced were chronicled. A password protected transcription software program was leveraged to transcribe voice recorded interview sessions. Taped interviews were transcribed verbatim, written transcripts were retrieved, isolated, grouped and regrouped for analysis. The sample size of 12 participants is small and therefore a limitation and made it difficult to generalize. The next chapter provides a deep dive into the results.

Chapter 4: Results

This chapter reports the analysis and coding of the data from 12 individual transcripts. Having explained adversity for the purpose of this study as negative experiences that have the possibility to disturb a person's adaptive functioning or development (Yates, Masten, 2004), adversity is something we all experience whether personally, professionally, or both.

Adversity comes in many forms, such as chronic, acute, systems within the individual, and the most common across participants at multiple level and settings simultaneously. As seen in Table 1, adversity can be experienced through major life disruptions, instability at home, parenthood, or manifest through more institutionalized experiences, such as school, work, or sports. Adversity has also been described as a teacher. For example, a CEO stated,

I have had the privilege of having a number of what I will call adverse circumstances. I think adversity is my absolute best teacher of all. I do not have to take notes about what I learned through adversity. It is burned into me like having a bad boss. I know what not to be because of him. Having a failed partnership, I know what not to do because of that failed partnership and it changed habit patterns. It changed brain patterns. And again, I did not have to work on it. I do not say it happened automatically because that's not how human change happens, but adversity has been my most, what is the right word, my most memorable teacher.

Adversity may also be a growth opportunity, as stated by another C-suite executive, "For me, adversity, I look at it as almost a growth opportunity. Like anytime there is a

challenge or struggle, specifically at work, whether it is with individuals [or] projects, I am kind of the first to the forefront to try and tackle it and to figure it out because I think that is how you learn." And finally, adversity may be seen as a catalyst as one Founder describes, "... adversity will push you into certain things and push you out of some things like getting laid off led me to move which... took my career in a whole other direction." For this sample, the most common type of adversity was multiple level and settings challenges simultaneously (e.g., broader systems of family, school, employment, etc.).

75% of the sample provided examples compared to only 25% of the sample identifying chronic type adversity (e.g., discrimination).

Table 1

Adversity

Theme	Exemplar Quote(s)	Participant Experience(s)
Chronic (e.g., poverty, discrimination)	"I am a Lesbian. I had to fight through the moments of it not being 'cool,' by being who I was and not being in the shadows of not being out. I have a wife and two kids and not caring what other people thought made me more successful."	3 (25%)
	"Growing up dirt poor made me want to do jobs that made me money and as I grew in those jobs and got the, what do you call it, the nest, I got the feeling of protection. I was able to move into positions that allow me to do work that was more creative, autonomous, and dynamic through my advocacy organization; projects that help people in a very real endeavor."	
	"I came from a blue-collar family as I think you can tell from my lack of polish. My Dad cleaned offices. Money was tight. I was in sixth grade and we would go help him with work right after school for like two hours. I would empty trash cans, clean the bathrooms and stuff like that. So, working in a white-collar office, man, it was like eye opening. It's like they've got free coffee! Maybe I was 11 when I realized there's something to aim for, it was like, that's what I want to do I want to work in an office."	
Acute (e.g., sudden loss of a loved one, addiction)	"I got sober when I was twenty-seven. For about ten years I was in and out of jail; I was an alcoholic and a drug addict. To get sober you have to be tired. For me it was either I get sober or I die."	2 (16%)
	"My daughter's death was like the ultimate failure for me. I failed at something that I cared more about than I cared about life. In my mind, in the early days, I felt like I had failed as a mother. How, how can I lose a daughter? You know, no good mother loses a daughter. I don't hold that to	

	be true anymore. That failure made it possible for me to be more human."	
Systems within the Individual (e.g., a virus that attacks the immune system)		0
Multiple Level and Settings Simultaneously (e.g., broader systems of family, school, employment, etc.)	"My parents didn't provide a stable environment through their separation and divorce; each parent got a child. They moved back and forth constantly. I moved twenty-one times by the age of seventeen. Every other weekend was on a plane and holidays at the airport."	9 (75%)
	"We moved around a lot. And so, I was constantly in this instability and not in a bad way, it wasn't scary or anything, my dad was in a position where he had to go and take over portfolios in different cities and whatnot. So, I was always having to make new friends and I got to learn to adapt to different environments and understand people and how they think, how they act like, you know, from a psychological standpoint too."	
	"Youth sports. Being put on stage in front of everyone. And feeling pressure for not just yourself, but others, like responsibility at a young age, and having others sort of witness it, there's no hiding if you failed. And so, I think that is the initial younger phase that then brought a more fearless person into the workplace."	
	"The school systems are designed to put you in a box. If you don't fit into the box and it doesn't work, then it's a struggle. Very early on I realized I'm on a different path. I didn't know exactly what that was, but I started to ask myself, what's open to me? The work wasn't challenging; it wasn't interesting to me. My mother and grandmother raised me together. They spent a lot of time telling me I was very smart; probably too much. It was difficult because your parents are dealing with the school system. School is not pleased with me, but I was in AP college level math classes while in seventh grade but failing every other class. It was by choice; it wasn't that I didn't understand	

English or History, it was just that I was completely uninterested. Adversity in a lot of ways feels like different problems."

"I was building a company and the partnership that I was creating failed. I would say it was more disruptive and upsetting than my divorce."

"I was born in Lithuania; my parents came to the states when I was really young. I was generally raised by American media because my parents were always working. Therefore, what I learned about the world and social norms, how I wanted to behave and act, came from watching American television because that was what I spent most my time doing because no one else had time to really spend time with me. So, I fell in love with the world of make believe."

"Leaving a secure job for a startup for pay that was roughly half of what I was making. My dad was like, dude, with the money you're making now, living at home, you could buy a house in a couple years, you can save money. But I had just gone through the end of a relationship, so it was kind of like a way to escape. At the end of the day, I kind of tapped into like passion and just said, hey, if I follow passion, then other things will kind of come in place, hopefully."

"Motherhood. I just don't give a s*** anymore. And because of that, I'm much more willing to take risks that I don't think I would have taken when I was younger."

This section focuses on career trajectory specifically individual career evolution and how experiences with adversity played a role in that evolution. Two key themes emerged in the interview transcripts and can be found in Table 2. First, half of the leaders in the sample described a hunger for learning and a key factor for leaving a job is when it is stifled. The second theme was the desire to and sense that they were making an impact in their current role.

Table 2

Career Trajectory: Evolution

Theme(s)	Exemplar Quote(s)	Participant Experience(s)
Learning	"What drives me is learning " if there's points in a role where you feel stuck and learning is stifled then those were the moments where I would look up and find that next thing."	6 (50%)
	"I have left jobs because my learning has actually started to plateau. So that's my goal is to make sure that I'm working on what is relevant, not just now, but what's going to be relevant, at least five years from now."	
	"If I find myself in a situation where I'm not learning a lot on a regular basis then it's time to go. I really need that. I strive to be filling my brain constantly. I believe you can learn even in bad situations. My challenge and impetus for moving on is not just improving on the things I already know but being exposed to a full spectrum so there's times where my creative side drives and my technology takes a backseat or vice versa."	
Impact	"Am I walking in and not excited about the impact I can make? What fuels me probably more than anything is being useful, being impactful. If I don't know what use I am, if I don't see my impact, what's in it for me and what's in it for the business? When I don't feel like I'm adding value anymore it begins to eat away at me, boredom, if it feels like it's deja vu - it's time to go."	4 (33%)
	"I honestly never would have thought I'd be in the position I am today. I knew I wanted to make an impact and a change whether it was industry or nonprofits."	

While two main themes dominated the responses, a few outliers were noticed as well. For example, a Founder stated,

I think I have just never been satisfied. I am sort of a perpetually dissatisfied soul and I am never content with the way things are, I always

kind of believe that they could be better. I think that there are many different trains that I could have gotten on along the way in my career and just said, this is good enough. And I never did because I always kept like shooting higher and wanting more.

Another CEO when asked about their career evolution and reason why they would leave a job said, "Ultimately, if I felt like my gifts and talents and abilities were not seen or engaged." This CEO's catalyst for departure was the lack of recognition and management's inability to tap into their strengths and further develop this individual's abilities.

The second focus of the career trajectory section is the ability to reach this level of career. The question, "Can you tell me a time when you leveraged a past experience with adversity to navigate the feeling of failure at work?" nearly always prompted a physical reaction whether a laugh or contorted face. However, the ability to use past adversity to address a current failure was not considered a key factor in the ability of these executives to reach their high positions. What is worthy to note is the commonalities across participants in their view of failure and the ability to rise above and press on. Failing is a part of a career trajectory. What is stands out across participants is the lack of acknowledgement, concern with it, and tendency to not dwell on it. One C-suite executive stated, "... this is where the hubris is actually kind of incredible. I genuinely cannot tell you a time where I think I failed at work. Minus failing to keep myself out of a situation or failing to see the signals and got too comfortable at a job at one point." While another Founder and CEO offered humor, "Did it ever dawn on you that I do not fail? No, I definitely fail but I guess that is it I do not think I fail. It is only

the next hustle, how do we move forward or how do I make believe that failure was intended? There is no time to wallow." The outcome was not that failure determined the ability to reach this level of their career instead it was categorized as perseverance, the ability to keep going. Table 3 outlines exemplar quotes for this career trajectory theme.

Table 3

Career Trajectory: Ability to Reach this Level of Career

Theme(s)	Exemplar Quote(s)	Participant Experience(s)
Positive Thinking	"Can do attitude; taking challenges as opportunities"	5 (42%)
	"Staying positive, I believe being at my low of the low of where I have been has made me who I am today. My adversity has made me see things from a grateful point of view. And I'm happy to be able to work hard and just to be there."	
	"Accepting where you are and recognizing it can't be worse than my worst time. No matter what I always ended up on my feet. I have been "ok" and believing no matter what you will be "ok" you reflect on well, if I've made it through that, I can make it through this."	
Vulnerability	"I think people respond well to people who are genuine and sincere and show humility, you know, and go to someone and say, I don't know how to do this, please help."	3 (25%)
Perseverance	"To me, failure is not making the calls. It's not trying."	7 (58%)
	"To us failing isn't an option. What's the saying "fail fast" I don't believe in that whatsoever. I think there are times you're going to fail but failures are successes at the end of the day because you take away so many learnings and a lot of times those projects turn into something new that is going to be, you know, a successful product."	

33

	"I can't remember specific failures. But I know that in my career "there's been plenty of them. I think you just like, learn your lesson, ingest it, and move on."	
Mentor	"I think two really big things for me were the value of having someone who did respect you and wanted to teach you like a good boss, a great influence, he made sure that we learned things, he made sure that you can come to him."	3 (25%)
	"I have an incredible mentor, somebody, who you know challenges me, has supported me from day one. And when I say support, it is not, you know, pat on the back you're doing a great job. I mean having my back when I do fail or make mistakes and then also going into solution mode instead of like, oh, s*** mode."	
Role Model	"Good role models. My grandmother was an executive at a company but came from nothing. When her parents brought her home from the hospital, they lived in a train car. She was from this incredibly conservative family. She was afraid of the changes but embraced them. Being able to see people do that while still having a family, living their lives, and doing things that matter, I think that shows you what is possible and that there are other paths available to you rather than just the normal lawyer, doctor, finance linear paths."	5 (42%)
	"I come from a long line of entrepreneurs. Somehow our family culture was 'you go out and make it happen.' I can remember my dad saying you can do anything you want to do, which, for a girl born in the early 60s was actually not true at all."	
	"I think it's witnessing my parents make the leap of faith, leaving their home country, where they had a comfortable lifestyle, but realizing that there's more available to them if they come to the States for their family long-term."	

Although five prominent themes emerged, there were a few outliers including

leveraging an ability to not care what others think as stated by one C-suite interviewee,

"To think I do not care allowed me to believe it. It worked better for me because I had watched so many executives be destroyed by the need to be liked and loved and affirmed and revered and I did not care about that or at least I did not think I did. It let take ownership of my career and not rely on or put my career in anyone else's hands because 'I do not care,' while maybe not healthy in my personal life, served me professionally because I did not let anyone else dictate my career. It only mattered what I cared about it.

Similar, but through a different lens, another C-suite executive reflected on their ability to not care what others think,

"Motherhood. I just do not give a s*** anymore. And because of that, I am much more willing to take risks that I do not think I would have taken when I was younger. So, I think there was a time in my life where those three things happened. Motherhood, just age, and hitting a point in my career where I just owned who I was a bit more and controlled my destiny a bit more.

Another Founder reflected specifically on leveraging their adversity,

Because I just have a working knowledge of a whole host of society and culture that many people do not have, I am able to utilize that and to figure out how to make that a positive. I have learned how to manipulate. As an example, I will give a speech and I know when to let my accent come out. I know when to make a self-deprecating comment. I know when to

reference, lets say, my tattoos for effect. And, so, it just gives me a bigger tool belt, which when wielded correctly, can be quite beneficial.

Table 4 focuses on the perception of leadership and how experiences with adversity have shaped their individual ways of leading. Two major themes emerged from the data. The first common theme and key factor in the view of leadership was the sense of responsibility. One C-suite executive reflected, "I once had 15 people reporting to me and that realization of the control you have over their lives and the daily effort that takes comes with a moment of what does this person see you as?" For another it was a moment of recognition that brought on the sense of responsibility,

I was given an award for one of the most influential leaders. Okay, I thought, so this is a group of people who I am not directly involved with, who are coming from outside and calling you a leader. Which means I am being seen by other people, which means the work I am doing is reaching other people. This may be because of my own issues, but I constantly need to be reminded I am a leader. And with that, I have responsibility.

And for another a moment of accomplishment, "After I made Partner for the first time, I perceived myself as a leader. I used to be the kid trying to get five or ten minutes of someone's time for advice. I now give the time wherever I can."

Table 4

Leadership

Theme	Exemplar Quote(s)	Participant Experience(s)
Responsibility	"Leadership to me is helping other people see their own potential but doing it in a way that I am absolutely relentless in holding them to it. That's how I lead. People listen to me and now I need to be responsible in where I take them."	7 (58%)
	"Fatherhood is when I perceived myself as a leader. I always talk about when you think about your children, we want to lead by example. And it's funny because you use the word lead. It's the same with work I wouldn't ask someone to do something that I wouldn't do myself."	
Empathy	"I don't judge people. I have a kinder heart. I realized you have to take the good from each person and learn from the bad in each of them. I took people for who there were."	5 (42%)
	"I try and really understand each person as an individual and not treat them like a group I oversee because everyone's incentivized and acts differently."	
	"I think that's the probably the most difficult part of leading, you have people dealing with change and struggling with those changes. And people don't like change and they want to come in and they want to know what's expected, and they want to know they can do their work. That has definitely shifted the way we approach our people. We're very conscious about taking our time when something goes wrong. We're very careful about approaching things in a very genuine way and making people realize that we're in it to make sure they have a good experience here."	

The second major theme, but almost equal in frequency with the first theme, was empathy. Empathy, for the majority, emerged as an outcome with direct correlation to the individual's experience with personal adversity. As one CEO states,

I have lived through the unimaginable, the suicide of a child, and I have lived to find joy, to live fully again. I am not a leader that has it all together and I do not pretend to, I do not need to. I am adamant that people take good care of their families. As a leader I have created a lot of room for care and nurture, not just for children but their aging parents, etc.

For another C-suite interviewee, it was their experience with another empathetic leader that instilled empathy as a core part of their leadership as shown by exemplar quote,

On September 11th, I was in my office, and the CEO at the time was walking by and he did not know who I was at that point in my career, but my name was somewhere visible. And he walked in and thanked me for being at work that day. And for me that was the moment where I was like, okay, it is not just about how do we drive decisions but how to do it with empathy?

Although there were two overarching themes, there were also outliers. One Founder reflected on how experiences with personal adversity still impacts them today, the ability to recognize it, and leverage those around them for support, "I still have that confrontational streak in me. And so, I kind of do the opposite of that. I send my employees if I am triggered or check-in with them on a response I have crafted. My leadership, my leadership style, empowering employees and checking myself with them."

Similarly, another C-suite reflects on their personal adversity and leverages their learnings from it,

I am not afraid to be vulnerable in front of my team which I have heard they really appreciate. I am honest and level set with them. I am not afraid to tell them things are bad. So therefore, they feel like they can trust me and are in the know. And I think that comes from my parents. When we were growing up and, you know, wanting to make ends meet my parents were always really real with me about the financial situation, the challenges, or if someone was unhealthy in the family, and they needed to deal with that.

From another viewpoint, a Partner discussed the difficulty of having a direct report leave because of their leadership style and that lesson learned, "He was a pretty experienced guy. And I did not give him enough room to run. I kind of boxed him in, and so I have since then tried to hire good people and let them operate their way and provide some guardrails, but not be rigid."

This last section shows the participants' ability to grow from adversity and the process they leverage to make it meaningful to them. Table 5 provides the reader with a deeper understanding of the individual actions that participants take for sense-making. There were three main themes with reflection being most prominent. One Partner discussed their own journey through reflection, "You got to learn from the past, but you got to let it go. If you always replay the past, you will always be fixing. A defining moment for me was when I was no longer living in the past."

Table 5

Adversity: The Process and Ability to Grow from It

Theme	Process	Participant Experience(s)
Therapy	Counseling session with a professional	3 (25%)
Reflection	Defined as serious thought or consideration	6 (50%)
	In writing (journaling, pro-con list, benchmarking)	3 (25%)
	Self-talk	2 (17%)
Reading		3 (25%)

Two outliers emerged with one Founder fondly connecting music as an outlet and reflective tool for processing and making adversity meaningful, while another Founder, relied heavily on their support system, "My wife, my sister, I am not the kind of person who can process things in my head, I need to talk to other people about it. I need to make sense about what I'm thinking about it through talking to someone else."

Summary

Each of the 12 participants had an experience with personal adversity in one form or another and many experienced more than one type. Nine participants identified a multiple level and settings simultaneously adverse experience while three participants identified a chronic version. Six participants (50%) noted that they had left a current employer or career if they felt their learning was stifled. Four themes emerged as the ability to reach this level of career: seven participants associated perseverance, five participants associated positive thinking, and eight participants recognized an external influence whether mentor or role model. Two key themes were linked to leadership, the

perception of and way of leading, seven participants associated a sense of responsibility with leadership and five participants identified empathy as a direct result of their experiences with personal adversity. Finally, reflection including serious thought, writing, or self-talk was the process most relied upon to make experiences with adversity meaningful to the participants.

Chapter 5: Discussion

This qualitative study utilized data from 12 interviews with high-level executives to add to the literature on leader and leadership development. Specifically, the interview data provided insights into the impact personal adversity had on the career trajectory and leadership styles of high-level executives, including their career evolution, perception of leadership, and how those experiences impacted their way of leading. Participants had various experiences with personal adversity and provided a deeper understanding of the types and severity levels of those experiences.

Career trajectory, the evolution and ability to reach this level in their careers, provided insights into the desire for continued learning and making an impact. Those two factors determined whether a leader would stay in their current role or seek out an opportunity that met that criteria. A combination of attributes and external forces were considered by leaders with regard to their ability to reach this level in their career. Positive thinking, vulnerability, perseverance, mentors and role models were identified as key factors in their trajectory. The impact of personal adversity on leadership resulted in responsibility and empathy as key themes drawn on from their experiences. Frequently referenced in the data was external acknowledgement of their leadership status that contributed to their perception of leadership, a strong moral code of treating others as you would like to be treated, and the desire to pay it forward. Finally, there is a process that emerged from the data to make sense of and grow from personal adversity identified through coping mechanisms and outlets with reflection being most prominent followed by therapy and reading.

The positive scholarship lens prompts researchers to expand focus from

describing what is problematic to capturing mechanisms that enable human flourishing. Positive scholarship represents a quest or desire to understand the processes that produce certain collective and individual states that are less commonly addressed by current organizational studies (Roberts, 2006).

Implications for Scholarship

Many participants referenced a childhood adverse event that had either influenced their way of leading or ability to reach this level in their career. The findings from this study provide support for the findings from the previous study conducted by Mittal et al. (2015). For example, if a person grows up in an unpredictable and constantly changing environment, he or she ought to develop cognitive tendencies that help him or her function adaptively in this type of challenging environment. In addition, Powell and Baker (2014) found that understanding the answer to the "why" question helps to explain much of the variation in *how* firms respond. The key step is less a matter of explicit decision making and more a matter of "bracketing," where some features of the adversity become pertinent and others are downplayed or ignored in creating the definition of the situation.

Little prior theory is available to help us to see or understand the application of adversity. This study raises questions around whether the level of severity and type of adversity determine specific outcomes, better understanding the application of adversity from childhood through adulthood may contribute to early therapy or training techniques that inform career advisors or enable human flourishing. To *flourish* means to live within an optimal range of human functioning, one that connotes goodness, generativity, growth, and resilience (Fredrickson & Loasada, 2005).

The findings of this study have important implications for current and future leaders, whether recent graduates entering the workforce or those looking to change or advance their career. This study achieves this by adding to the conversation of how we portray leaders and what makes up a leader profile; it is not one size fits all. Participants leveraged learnings from experiences with personal adversity and applied them professionally. Empathy and responsibility were noted as key factors that shaped their individual view of leadership and way of leading. Leaders need to manage emotions such that followers are aware of problems yet, given the collective vision, are confident about resolving problems and feel optimistic about the efficacy of their personal contribution. Not only do leaders have to meet these multiple demands, but they also have to constructively resolve conflicts, and generate and maintain a sense of cooperation and trust (George, 2000).

The leadership literature has typically focused on trauma inside the workplace and the behaviors and tools associated to overcome those hardships, such as abusive supervision and harassment. The corrupt executives at institutions such as Enron, Worldcom, and Tyco were not mere symbols of corporate greed and malfeasance. Bad leadership at Enron alone impoverished thousands of employees, stealing their livelihoods, gutting their retirement accounts, and tearing them apart with stress (Bennis, 2007). This study addressed a gap in the literature as it relates to the application of experiences with personal adversity that were had prior to or in tandem with the hardships at work.

Implications for Practice

This study investigated the career trajectory of leaders, specifically development

and the ability to reach this level in their career. The desire for learning and impact are interesting data points for those seeking career opportunities and organizations that are looking to retain and cultivate top talent. Candidates applying for roles can interview for the appetite to provide learning and impact within an organization or role. It can also be systemically built into an organization's culture through flexible job descriptions, crossfunctional skill building, project work opportunities, and implemented by leadership and management.

While conducting interviews, some participants asked how to identify profiles similar to themselves, self-motivated, capable, dedicated, and eager to advance. Similar to resilience training, the themes of positive thinking, vulnerability, and perseverance, traits identified by the leaders with regard to their ability to reach this level in their career, are teachable. Shin et al. (2012) argued that positive affect is a crucial mediator of the effects of psychological resilience upon employees' commitment to change. This relationship results from resilient individuals' understanding of the value of positive emotions and their skills in evoking them (e.g., using their sense of humor and developing effective relaxation techniques), as well as from their possession of coping resources to keep negative emotions under control. These findings are supported by participant's offering humor and an inability to "see" failure and with the data showing defaulting to reflection through writing, self-talk or reading, and therapy as healthy coping techniques. These attributes can also be interviewed for with focused questions around experiences with failure and characteristics leveraged to solve problems and cope with challenges.

In addition, positive thinking, vulnerability, and perseverance can be more prominent in our education system and reinforced by our educators and coaches. Mentorship and role models were external themes identified for the ability to reach this level of career. There is significant research on the positive influence of mentors and role models from forming an occupational identity as a young adult to reassessment of midcareer/midlife, as demonstrated by Kram's (1983) research the mentor relationship has great potential to facilitate career advancement and significantly enhance psychological development both in early and middle adulthood. Given the impact of these findings in support of leaders continued development additional research around mentorship requirements, criteria, and accessibility may be valuable. For example, the last year of high school is a formative time and the readiness for the next step is in question, this may be an impactful time for additional support.

Prescod and Zeligman (2018) found mental health and career counseling, two disciplines that are often discussed as separate entities, can be beneficially integrated. Consistent with Prescod and Zeligman (2018), there may be opportunities for future studies to be done with regard to mental health and the level to which individuals know themselves, make sense of experiences, and then identify careers to pursue. Mental health also has a role in one's ability to not only cope with but thrive from adverse experiences. Leaders deferred to reflection, therapy, and reading as their process. Effective counselors will infuse work on coping skills into their career counseling to ensure clients have healthy coping skills for moving forward in all areas of their lives, including vocational areas (Grant, 2014).

Limitations

Interviewing poses strengths and limitations. Interviewing provides indirect information filtered through the views of interviewees (Creswell, 2014). Limitations include small sample size, overwhelming amount of data that can be difficult to analyze and absorb making generalizations difficult, and varying levels of articulation. The researcher's presence may bias responses (Creswell, 2014). There is potential for researcher bias as participant's were recruited from the researcher's personal network.

Conclusion

The data provided a bird's eye view of leaders emerging from different experiences with personal adversity. The level of severity and type of adversity fluctuated however the data remained mostly consistent across the varying levels. The ability to acknowledge, reflect, make sense of, and grow from their personal adversity is what is unique in each of these leaders. The lack of concern for failure and in some instances an ability to disregard other opinions when it did not serve them is a testament to the value of leveraging learnings from personal adversity and its application professionally.

I believe as a society we can do better by introducing healthy coping mechanisms and processes for making sense of our experiences at much earlier stages in education, including as early as preschool. This study provided a deeper understanding of the impact personal adversity had on the career trajectory and leadership among high-level executives. The data was developed using a narrative approach to interviews with high-level executives. Finally, important implications from these interviews were discussed with reference to leadership in regard to personal adversity, profile, and career trajectory as a subject of both research and personal development.

References

- Aldwin, C. M. & Levenson, M. R. (2004). Posttraumatic Growth: A Development Perspective. *Psychological Inquiry*, *15*(1), 19-92.
- Anthony, E. L. (2017). The impact of leadership coaching on leadership behaviors. *Journal of Management Development*, *36* (7), 930-939. https://doi.org/10.1108/JMD-06-2016-0092
- Avolio, B. J., Walumbwa, F. O., & Weber, T. J., (2009). Leadership: Current Theories, Research, and Future Directions. *Annual Review of Psychology*, 60, 421-449.
- Bartone, P.T., Kelly, D. R., & Matthews, M. D. (2013). Psychological Hardiness Predicts Adaptability in Military Leaders: A prospective study. *International Journal of Selections and Assessment*, 21 (2), 200-210.
- Bear, J. B., Todorova. G., & Weingart, L. R. (2014). Can Conflict Be Energizing? A Study of Task Conflict, Positive Emotions, and Job Satisfaction. *Journal of Applied Psychology*, 99(3), 451-467.
- Bennis, W. (2007). The Challenges of Leadership in the Modern World. *American Psychologist*, 62(1), 2–5. DOI: 10.1037/0003-066X.62.1.2
- Bennis, W. & Thomas, R. (2002). Geeks & geezers: How era, values, and defining moments shape leaders. Boston, MS: Harvard Business School Press.
- Bennis, W. G., & Thomas, R. J. (2002). Crucibles of Leadership. *Harvard Business Review*, 80(9), 39-45.
- Betancourt, T. S., Borisova, I., Williams, T. P., Meyers-Ohki, S. E., Rubin-Smith, J. E., Annan, J., & Kohrt, B. S. (2013a). Research review: Psychosocial adjustment and mental health in former child soldiers-A systematic review of the literature and recommendations for future research. *Journal of Child Psychology and Psychiatry*, *54*(1), 17-36. Doi: http://dx.doi.org/10.1111/j.1469-8624.2012.01861.x
- Bonanno, G.A. (2004). Loss, Trauma, and Human Resilience: Have We Underestimated the Human Capacity to Thrive After Extremely Aversive Events? American Psychological Association, Inc. 0003-066X/04/59(1), 20–28 DOI: 10.1037/0003-066X.59.1.20
- Bonanno, G.A. (2005). Clarifying and Extending the Construct of Adult Resilience. *American Psychologist*, 60(3), 265–267. DOI: 10.1037/0003-066X.60.3.265b

- Bonanno, G. A.& Mancini, A. D. (2006). Resilience in the Face of Potential Trauma: Clinical Practices and Illustrations. *Journal of Clinical Psychology*, 62(8), 971-985
- Bonanno, G. A., Romero, S. A., & Klein, S. I. (2015). The Temporal Elements of Psychological Resilience: An Integrative Framework for the Study of Individuals, Families, and Communities. *Psychological Inquiry*, 26, 139-169. DOI: 10.1080/1047840X.2015.992677
- Bernstein, S. D. (2003). Positive Organizational Scholarship: Meet the Movement. An Interview with Kim Cameron, Jane Dutton, and Robert Quinn. *Journal of Management Inquiry*, 12(3), 266-271. DOI: 10.1177/1056492603256341
- Bright, D. S. (2009). Appreciative Inquiry and Positive Organizational Scholarship: A Philosophy of Practice for Turbulent Times. OD Practitioner, Vol. 41, No. 3
- Bright, D. S., Powley, E. H., Fru, R. E., & Barrett, F. (2013). The generative potential in crucial conversations. *Advances in Appreciative Inquiry*, *4*, 135157. Emerald Group Publishing Limited. ISSN: 1475-9152/doi:10.1108/S1475-9152(2013)0000004005
- Byron, K., Myrowitz, J., Peterson, S. J., & Walumbwa, F.O. (2009). CEO Positive Psychological Traits, Transformational Leadership, and Firm Performance in High-Technology Start-up and Established Firms. *Journal of Management*, *35*(2), 348-368.
- Calhoun, L.G. & Tedeschi, R. G. (2004). The Foundations of Posttraumatic Growth: New Considerations. *Psychological Inquiry*, *15*(1), 93-102.
- Calhoun, L.G. & Tedeschi, R. G. (2004). Posttraumatic Growth: Conceptual Foundations and Empirical Evidence. *Psychological Inquiry*, 15(1), 1-18.
- Cicchetti, D. (2010). Resilience under conditions of extreme stress: A multilevel perspective. *World Psychiatry*, *9*, 145-154.
- Creswell, J. W. (2013). Research design: Qualitative, quantitative, and mixed methods approaches. Thousand Oaks, CA: Sage.
- DiClementi, G., Levitt, K., & McClellan, J. (2017). Emotional Intelligence and Positive Organizational Leadership: A Conceptual Model for Positive Emotional Influence. *Journal of Behavioral and Applied Management*, 17(3), 197-212.
- Dimas, I. D., Rebelo, T., Lourenço, P. R., Isabel, Pessoa, C. I. P. (2018) Bouncing Back from Setbacks: On the Mediating Role of Team Resilience in the Relationship Between Transformational Leadership and Team Effectiveness. *The Journal of Psychology*, *152*(6), 358-372, DOI: 10.1080/00223980.2018.1465022

- Duchek, S. (2017). Entrepreneurial resilience: a biographical analysis of successful entrepreneurs. *International Entrepreneurship and Management Journal*, *14*, 429-455. DOI: 10.1007/s11365-017-04467-2
- Fineman, S. (2006). On Being Positive: Concerns and Counterpoints. *Academy of Management Review*, 31(2), 270-291.
- Fredrickson, B.L. (2005). Positive Affect and the Complex Dynamics of Human Flourishing. *American Psychologist*, 60(7), 678-686, DOI: 10.1037/0003-066X.60.7.678
- Fredrickson, B. L., Cohn, M. A., Coffey, K. A., Pek, J., & Finkel, S. M. (2008). Open Hearts Build Lives: Positive Emotions, Induced Through Loving-Kindness Meditation, Build Consequential Personal Resources. *Journal of Personality and Social Psychology*, *95*(5), 1045-1062. DOI: 10.1037/a0013262
- Fredrickson, B. L. & Tugade, M. M. (2004). Resilient Individuals Use Positive Emotions to Bounce Back From Negative Emotional Experiences. *Journal of Personality and Social Psychology*, 86(2), 320-333.
- Fredrickson, B. L., Tugade, M. M., Waugh, C. E., & Larkin, G. R. (2003). What Good Are Positive Emotions in Crises? A Prospective Study of Resilience and Emotions Following the Terrorists Attacks on the United States on September 11th, 2001. *Journal of Personality and Social Psychology*, 84(2), 365-376. DOI: 10.1037/0022-3514.84.2.365
- Fugate, M. E. L., Kinicki, A. J., & Prussia, G. E. (2008). Employee coping with organizational change: an examination of alternative theoretical perspectives and models. *Personnel Psychology*, *61*(1), 1-36.
- George, J. M. (2000). Emotions and Leadership: The Role of Emotional Intelligence. *Human Relations*, *53*(8), 1027–1055. https://doi.org/10.1177/0018726700538001
- Grant, A. M. (2014). The Efficacy of Executive Coaching in Time of Organisational Change. *Journal of Change Management*, *14*(2), 258-280. http://dx.doi.org/10.1080/14697017.2013.805159
- Huy, Q. N. (1999). Emotional capability, emotional intelligence, and radical change. *Academy of Management Review*, 24, 325-345.
- Janoff-Bulman, R. (2004). Posttraumatic Growth: Three Explanatory Models. *Psychological Inquiry*, *15*, 30-34.
- Kirkpatrick, S. A. & Locke, E. A. (1991). Leadership: do traits matter? *Academy of Management Executive*, 5(2), 48-60.

- Kramer, C. A. & Allen, S. A. (2018). Transformational Leadership Styles Pre- and Post-Trauma. Journal of Leadership Education. DOI: 10.12806/V17/13/R5
- Luthans, F. & Youssef, C.M. (2007). Positive Organizational Behavior in the Workplace, The Impact of Hope, Optimism, and Resilience. *Journal of Management*, *33*(5), 774-800. DOI: 10.1177/0149206307305562
- Luthar, S. S. (2006). Resilience in development: a synthesis of research across five decades. In D. Cicchetti & D.J. Cohen (Eds.), *Developmental psychopathology*, *3*, Rick, disorder, and adaptation (2nd ed., pp. 739-795). Hoboken, NJ: Wiley.
- Maggiori, C., Johnston, C. S., Krings, F., Massoudi, K., & Rossier, J. (2013). The role of career adaptability and work conditions on general and professional well-being. *Journal of Vocational Behavior*, 83, 437–449. doi:10.1016/j.jvb.2013.07.007
- McClellan, J, Levitt, K., & DiClementi, G. (2017). *Journal of Behavioral & Applied Management*, 17(3), 197-212.
- Masten, A. S. (2001). Ordinary Magic: Resilience Processes in Development. *American Psychologist*, 56(3), 227-238. DOI: 10.1037//0003-066X.56.3.227
- Masten, A. S. (2015). Pathways to Integrated Resilience Science. *Psychological Inquiry*, 26, 187-196. DOI: 10.1081/1047840X.2015.1012041
- Maxwell, J. A. (2013). Qualitative Research Design: An Interactive Approach. SAGE Publications, Inc, Edition 3.
- Mittal, C., Griskevicius, V., Simpson, J. A., Sooyeon, S., & Young, E. S. (2015). Cognitive Adaptations to Stressful Environments: When Childhood Adversity Enhances Adult Executive Function. *Journal of Personality and Social Psychology*, *109*(4), 604-621. http://dx.doi.org/10.1037/pspi0000028
- Moldoveanu, M. & Narayandas, D. (2019). The Future of Leadership Development. Harvard Business Review. Retrieved from https://hbr.org/2019/03/educating-the-next-generation-of-leaders
- Ovans, A. (2015). What Resilience Means and Why It Matters. Harvard Business Review. Retrieved from https://hbr.org/2015/01/what-resilience-means-and-why-it-matters
- Overland, G. (2011). Generating theory, biographical accounts and translation: a study of trauma and resilience. *International Journal of Social Research Methodology*, 14(1), 61-75.

- Powell, E. E. & Baker, T. (2014). It's What You Make of It: Founder Identity And Enacting Strategic Responses To Adversity. *Academy of Management Journal*, 57(5), 1406-1433. http://dx.doi.org/20-5465/amj.2012.0454
- Prescod, D. J., Zligman, M. (2018). Career Adaptability of Trauma Survivors: The Moderating Role of Posttraumatic Growth. *The Career Development Quarterly*, 66, 107-120.
- Rendon, J. (2015, July). How Trauma Can Change You—For the Better. Time, Retrieved from http://time.com/3967885/how-trauma-can-change-you-for-the-better/
- Roberts, L. M. (2006). Shifting The Lens On Organizational Life: The Added Value of Positive Scholarship. *Academy of Management Review*, 21(2), 292-305.
- Shin, J., Taylor, M. S., & Seo, M. G. (2012). Resources for change: The relationships of organizational inducements and psychological resilience to employees' attitudes and behaviors toward organizational change. *Academy of Management Journal*, 55(3), 727–748.
- Tedeschi, R. G.; Calhoun, L.G. (2004). Posttraumatic Growth: Conceptual Foundations and Empirical Evidence. *Psychological Inquiry*. *15*(1), 1-18. DOI: 10.1207/s15327965pli1501_01
- Tedeschi, R. G., Park, C. L., & Calhoun, L.G. (1998). Posttraumatic Growth: Positive Changes in the Aftermath of Crisis. Mahwah, New Jersey. Lawrence Erlbaum Associates, Publishers.
- Teo, W.L., Lee, M., & Lim, W.S. (2017). The relational activation of resilience model: How leadership activates resilience in an organizational crisis. *Journal of Contingencies and Crisis Management*, 25, 136-147. https://doi.org/10.1111/1468-5973.12179
- Tugade, M.M., Fredrickson, B.L. (2004). Resilient Individuals Use Positive Emotions to Bounce Back From Negative Emotional Experiences. *Journal of Personality and Social Psychology*, 86(2), 320-333. DOI: 10.1037/0022-3514.86.2.320
- Yates, T. M., & Masten, A. S. (2004). Fostering the Future: Resilience theory and the practice of positive psychology. Hoboken, NJ: Wiley and Sons.
- Youssef, C. M. & Luthans, F. (2007). Positive Organizational Behavior in the Workplace The Impact of Hope, Optimism, and Resilience. *Journal of Management*, *33*(5), 774-800. DOI: 10.1177/0149206307305562
- Williams, J. & Allen, S. (2015). Trauma-Inspired Prosocial Leadership Development. *Journal of Leadership Education*, 14(3), 86-103.

Znoj, H. (2006). *Bereavement and Posttraumatic Growth*. In L. G. Calhoun & R. G. Tedeschi (Eds.), *Handbook of posttraumatic growth: Research & practice* (p. 176–196). Lawrence Erlbaum Associates Publishers.

Appendix A: Informed Consent

PEPPERDINE UNIVERSITY

Graziadio School of Business and Management

INFORMED CONSENT FOR PARTICIPATION IN RESEARCH ACTIVITIES

Leadership and Career Trajectory: The Role of Personal Adversity

You are invited to participate in a research study conducted by Kristina Kovacs, MSOD graduate student under the supervision of faculty advisor, Darren Good, Ph.D. at Pepperdine University, because you are a C-suite, President, Partner or Founder. Your participation is voluntary. You should read the information below, and ask questions about anything that you do not understand, before deciding whether to participate. Please take as much time as you need to read the consent form. You may also decide to discuss participation with your family or friends. You will also be given a copy of this form for your records.

PURPOSE OF THE STUDY

The purpose of this study is to understand the role personal adversity may have in supporting the careers and leadership among high-level executives. The objective is to better understand the characteristics, factors, and / or behaviors associated with personal adversity that contributed to leaders' perceived career growth and leadership style.

STUDY PROCEDURES

If you volunteer to participate in this study, you will be asked to participate in a 45-minute in-depth audio recorded interview either face-to-face or via skype depending on your availability. You will be instructed not to identify your name or industry during anytime of the recorded interview. The recordings will be encrypted, housed on the principal investigator's G-suite drive so no one except the principal investigator can access the information. The interviews will be voice-recorded and transcribed. Any transcribed or handwritten notes will be analyzed and coded based on emerging themes and patterns.

POTENTIAL RISKS AND DISCOMFORTS

The potential and foreseeable risks associated with participation in this study are minimal. Identity will remain anonymous and you will be subject to the same emotional reactions you may have during typical conversation while discussing an incident with adversity. To mitigate any psychological discomfort, you will be allowed to pause and

resume at your will. The principal investigator will focus on how the incident affects you today rather than reliving any prior experiences with adversity.

POTENTIAL BENEFITS TO PARTICIPANTS AND/OR TO SOCIETY

While there are no direct benefits to the study participants, there are several anticipated benefits to society which include adding to existing literature on leadership, resilience, and positive organizational scholarship. Additionally, there are opportunities to create programing, training and/ or coaching for current and future leaders by identifying and understanding the characteristics, factors, and/ or behaviors associated with personal adversity that contributed to leaders' perceived career growth and leadership.

CONFIDENTIALITY

The records collected for this study will be anonymous far as permitted by law. However, if required to do so by law, it may be necessary to disclose information collected about you. Examples of the types of issues that would require me to break confidentiality are if disclosed any instances of child abuse and elder abuse. Pepperdine's University's Human Subjects Protection Program (HSPP) may also access the data collected. The HSPP occasionally reviews and monitors research studies to protect the rights and welfare of research subjects.

There will be no identifiable information obtained in connection with this study. Your name, address, or other identifiable information will not be collected. The interviews will be voice-recorded. The data collected will be encrypted, housed on the principal investigator's G-suite drive so no one except the principal investigator can access the information. Any transcribed or handwritten notes will be kept in a locked file case in the principal investigator's residence. The data will be stored on a password protected computer at the researcher's place of residence for three years after the study has been completed and then destroyed.

SUSPECTED NEGLECT OR ABUSE OF CHILDREN

Under California law, the researcher(s) who may also be a mandated reporter will not maintain

as confidential, information about known or reasonably suspected incidents of abuse or neglect

of a child, dependent adult or elder, including, but not limited to, physical, sexual, emotional, and

financial abuse or neglect. If any researcher has or is given such information, he or she is required to report this abuse to the proper authorities.

PARTICIPATION AND WITHDRAWAL

Your participation is voluntary. Your refusal to participate will involve no penalty or loss of benefits to which you are otherwise entitled. You may withdraw your consent at any

time and discontinue participation without penalty. You are not waiving any legal claims, rights or remedies because of your participation in this research study.

ALTERNATIVES TO FULL PARTICIPATION

The alternative to participation in the study is not participating.

INVESTIGATOR'S CONTACT INFORMATION

You understand that the investigator is willing to answer any inquiries you may have concerning the research herein described. You understand that you may contact Kristina Kovacs at Kristina.Kovacs@pepperdine.edu or Darren Good, Ph.D. at Darren.Good@pepperdine.edu if you have any other questions or concerns about this research.

RIGHTS OF RESEARCH PARTICIPANT – IRB CONTACT INFORMATION

If you have questions, concerns or complaints about your rights as a research participant or research in general please contact Dr. Judy Ho, Chairperson of the Graduate & Professional Schools Institutional Review Board at Pepperdine University 6100 Center Drive Suite 500 Los Angeles, CA 90045, 310-568-5753 or gpsirb@pepperdine.edu.

Appendix B: Interview Script

<u>Purpose of the study</u>: The role personal adversity may have in supporting the careers and leadership among high-level executives.

Main Research Questions:

- 1. How have experiences with adversity affected your career trajectory?
- 2. How have experiences with adversity affected you as a leader?
- 3. To what do you attribute your ability to grow from adversity?

Follow-up Interview Questions:

Addressing Main Research Question 1:

- 1. Can you tell me about an experience with adversity that influenced your thoughts on what you wanted to do when you grew up?
- 2. How has your career evolved over time?
 - 2a. Did experiences with adversity play a role in that evolution?
- 3. To what do you attribute your ability to reach this level in your career? 3a. Was it an experience? A person?
- 4. Can you tell me a time when you leveraged a past experience with adversity to navigate the feeling of failure at work?

Addressing Main Research Question 2:

- 1. Can you pinpoint the moment when you experienced the perception of yourself as a leader?
 - 1a. What was happening? How were others responding to you? What behaviors were on display? How did you feel?
- 2. Can you give a specific example of how experiences with adversity impact your leadership behavior today?
- 3. How have experiences with adversity impacted your business practices?

 3a. Setting the mission/vision/values, storytelling, or strategy

Addressing Main Research Question 3:

- 1. To what do you attribute your ability to grow from adversity?1a. What was your process? E.g. Did you seek help? If so, what kind? For how long? If not, what did you turn to?
- 2. What did you do in your process to make adversity meaningful to you?2a. Do you leverage this process today? If so, when? Under what circumstances?Can you give me an example?